Author's response to reviews

Title: Training in Dual Diagnosis Interventions (The COMO Study): A Randomised Controlled Trial

Authors:

Elizabeth C Hughes (lhughes@lincoln.ac.uk)
Shamil Wanigaratne (shamil.wanigaratne@slam.nhs.uk)
Kevin Gournay (keyingournay@aol.com)
Sonia Johnson (s.johnson@ucl.ac.uk)
Graham Thornicroft (graham.thornicroft@iop.kcl.ac.uk)
Emily Finch (emily.finch@slam.nhs.uk)
Jane Marshall (jane.marshall@slam.nhs.uk)
Neil Smith (n.smith@iop.kcl.ac.uk)

Version: 7 Date: 23 January 2008

Author's response to reviews:

Added trial registration number and deleted table references and figure 1